

Format for ANSWERING REVIEWERS

December 22, 2014

Dear Editor,

Please find in the online submission system the edited manuscript in Word format.

Title: Anesthesia and Acupuncture

Author: Gerhard Litscher, Holger Simonis, Wolfgang Kröll

Name of Journal: *World Journal of Anesthesiology*

ESPS Manuscript NO: 15061

The manuscript has been improved according to the suggestions of reviewers:

Revision has been made according to the suggestions of the reviewer 01076129

- (1) We have included all aspects stated in this reviewer's first ten lines in our introduction.
- (2) We have deleted the sentence in question.
- (3) The paragraph in question has also been deleted.
- (4) We have added the reviewer's last sentence in our conclusion.

Revision has been made according to the suggestions of the reviewer 02484487

- (1) As requested by the reviewer, we have added an objective.
- (2) We have intensified the research on the topic and added several more references to our list.
- (3) We have given some strong points (in our opinion) as suggested by the reviewer.
- (4) We have structured our manuscript as requested.

The last reviewer said "No changes are needed".

Thank you again for publishing our manuscript in the *World Journal of Anesthesiology*.

Sincerely yours,



Gerhard LITSCHER

Prof. MSc PhD MDsc

Head of the Research Unit of Biomedical Engineering
in Anesthesia and Intensive Care Medicine and of the
TCM Research Center Graz

MEDICAL UNIVERSITY GRAZ

Auenbruggerplatz 29, A-8036 Graz/Austria

Tel: +43 316 385-13907 Fax: +43 316 385-13908

Gerhard LITSCHER, MSc PhD MDsc

Research Unit of Biomedical Engineering in Anesthesia and Intensive Care Medicine and TCM
Research Center Graz

Medical University of Graz

Auenbruggerplatz 29

8036 Graz, Austria

Tel. +43 316 385-13907, Fax: +43 316 385-13908

E-mail: gerhard.litscher@medunigraz.at

